Author's response to reviews

Title: California Men's Health Study (CMHS): A multiethnic cohort in a managed care setting

Authors:

Shelley M Enger (shelleyenger@yahoo.com)
Stephen K Van Den Eeden (Stephen.Vandeneeden@kp.org)
Barbara Sternfeld (Barbara.Sternfeld@kp.org)
Ronald K Loo (Ronald.K.Loo@kp.org)
Charles P Quesenberry Jr (Charles.Quesenberry@kp.org)
Sarah Rowell (Sarah.Rowell@kp.org)
Marianne C Sadler (Marianne.C.Sadler@kp.org)
Donna M Schaffer (Donna.M.Schaffer@kp.org)
Laurel A Habel (Laurel.Habel@kp.org)
Bette J Caan (Bette.Caan@kp.org)

Version: 3 Date: 9 June 2006

Author's response to reviews:

Per the email of 5/31/06, we have revised the manuscript to meet the requirements stated. In summary, changes to the title, author list, Affiliations, Running head, Abstract, Keywords, References, Section headings, and figure and table placement notation were made (either changed, or eliminated, per instructions).

Thank you for publishing our paper.